RESEARCH

Open Access

Enhancing quality of life and medication adherence for people living with HIV: the impact of an information system

Maria Panayi^{1,2}, Georgios K. Charalambous^{1,3} and Eleni Jelastopulu^{1,4*}

Abstract

Background The widespread availability of antiretroviral therapy has led to improvements in life expectancy and thus an increase in the number of people living with HIV/AIDS (PLWHA) worldwide. However, a similar increase in the number of newly-diagnosed patients in Cyprus suggests the need for solutions designed to improve monitoring, planning, and patient communication. In this study, we aimed to determine whether the use of an information system to manage PLWHA might contribute to improved quality of life and critical adherence to prescribed drug regimens and ongoing medical care.

Methods A randomized controlled trial study was conducted in Cyprus based on information that we collected using the highly valid and reliable Greek translation of the World Health Organization (WHO) Quality of Life (QOL) HIV-BREF questionnaire to assess sociodemographic variables and patient compliance. We distributed 200 questionnaires before implementing a Health Medical Care (HMC) information system at our clinic. Six months after implementing this system, 68 of the completed questionnaires were selected, including two groups of 34 participants who had been assigned at random to the intervention or the control group. Participants included PLWHA aged ≥ 18 years who had been receiving antiretroviral therapy for more than 12 months between July 15, 2020, and July 15, 2022.

Results The changes in baseline to six-month scores reported for the intervention group were significantly higher than in the control group in all six subscales assessed with the WHOQOL-HIV-BREF questionnaire, as well as in the assessment of compliance. Furthermore, compliance with treatment was associated with higher scores in the questionnaire subscales, including physical health, psychological health, degree of autonomy, social relationships, life circumstances, and spirituality/religious/personal beliefs. We also identified specific demographic factors and behaviors that were associated with better compliance with scheduled medical care and the prescribed drug regimen. Specifically, men exhibited better compliance than women and younger PLWHA exhibited better compliance than the elderly as did individuals who reported a higher level of educational attainment. Additionally, individuals who did not use addictive substances, consumed less alcohol, and were managed using the monitoring information system all exhibited better compliance compared to those in the control group.

Conclusion The results of this study suggest that management of PLWHA via the use of an information system can contribute to improved QOL and drug compliance.

Keywords Adherence, Compliance, Quality of life, HIV/AIDS, Mobile health

*Correspondence:

Eleni Jelastopulu jelasto@upatras.gr

Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

Plain English Summary

This study was conducted to find ways to improve the lives of people living with HIV/AIDS (PLWHA) in Cyprus. With more PLWHA receiving treatment and care, it's crucial to ensure they have a good quality of life and follow their treatment plans. The main question we wanted to answer was whether using a special computer system to manage PLWHA could make their lives better and help them stick to their treatment. We found that using this computer system indeed made a positive difference. People who used it reported better physical and mental health, more independence, better relationships, and overall improved life circumstances. The study showed that managing PLWHA with a computer system can lead to better quality of life and treatment adherence. Some groups, like men, younger individuals, and those with more education, did even better. Also, people who didn't use addictive substances or consume much alcohol had better results. In summary, using technology in healthcare can greatly benefit PLWHA in Cyprus. This study highlights the importance of innovative solutions in healthcare to enhance the well-being of people living with HIV/AIDS and ensure they receive the best possible care.

Background

In light of extraordinary medical progress in the past several decades, several diseases that were once considered fatal have now become chronic conditions. This is clearly the case for people living with HIV/AIDS (PLWHA). With proper adherence to medication, PLWHA can experience prolonged life expectancy and a reasonable quality of life (QOL). According to the statistics from the United Nations for 2020, there are 37.7 million PLWHA worldwide, including 1.7 million children aged 0-14 years. Of note, 53% percent of the PLWHA are females. While 83% of all PLWHA are aware of their HIV-positive status, 6.1 million people worldwide remain unaware of their diagnosis. Furthermore, HIV/AIDS-related deaths have decreased by 64% since 2004 and by 47% since 2010. Only 680,000 HIV-associated deaths were reported in 2020, representing a significant decline in HIV-related mortality compared to what was observed in previous years. These positive developments have been attributed to the availability of effective antiretroviral therapy (ART) and critical patient education. While only 7.8 million PLWHA had access to treatment in 2010, by 2020, this number had increased to 27.5 million, with 74% of adults aged 15 years and older having access to ART. Additionally, a full 85% of pregnant women diagnosed with HIV had access to ART and could thus prevent transmission of the virus to their unborn children [1].

Nowadays, medical professionals globally are focused on achieving the WHO's ambitious goal, known as the "95%–95%–95%" target. This goal encompasses three key objectives designed to ensure that (i) 95% of the HIVinfected population is diagnosed, (ii) 95% of diagnosed individuals initiate antiretroviral therapy (ART), and (iii) 95% of those on ART achieve the desired outcome of an undetectable viral load [2]. As of 2020, among PLWHA who were aware of their HIV status, 87% had access to treatment, and 90% successfully achieved virus suppression. More specifically, among all PLWHA, 73% had initiated treatment and 66% had achieved viral suppression [1]. Unfortunately, the number of new HIV/AIDS cases continues to increase across the globe [3]. Thus, all countries must determine how to respond effectively to the growing demand for long-term care. Any decision or collective efforts directed at providing care for PLWHA should prioritize and strive for optimal QOL.

According to the new registrations of HIV-positive people in the system of the Republic of Cyprus, in the last ten years the registrations of people living with HIV have more than doubled. Specifically, from 1986 to 2010, 681 people with HIV were monitored, while from 2011 to 2021, the number of new registrations of people monitored at the Gregorios Clinic of Larnaca General Hospital, the only HIV National Reference center in Cyprus, amounts to 900 (Fig. 1) [4]. If we compare the last two decades, we can see that from 2000-2010 new registrations amounted to 274 while from 2011 to 2021 they have almost tripled. One of the most important problems in the management of these individuals is the constant increase in new cases (Fig. 1) [4]. This problem, combined with shortages in the medical and nursing staff and the increase of comorbidities due to long life expectancy [2, 4], makes proper and effective follow-up check of people living with HIV in Cyprus even more difficult.

The objective of this study was to determine whether managing PLWHA using a Health Medical Care (HMC) information system could have a positive impact on patient QOL and compliance. Our implementation and evaluation of a newly-established information system will permit new goals to be identified and established and may lead to improved monitoring, treatment planning, and ultimately a better QOL for PLWHA in Cyprus.

Methods

Study design-setting

This randomized controlled trial (RCT) study was conducted at the Gregorios Clinic of the Lanarca General



Fig. 1 Total number of new HIV/AIDS cases in Cyprus per year and gender and total prevalent cases, 2000–2022 (Epidemiological Bulletin of the Ministry of Health, Cyprus)

Hospital, which is the HIV National Reference clinic in Cyprus. Data collection took place between July 15, 2020 and July 15, 2022. Participants were selected randomly and completed the questionnaires during their routine visits to the Gregorios Clinic of Larnaca General Hospital.

Participants

The participants of this study were PLWHA who were monitored at the Gregorios Clinic of Larnaca General Hospital. Patients were included if they fulfilled the following criteria: $(1) \ge 18$ years of age; (2) actively monitored at the Gregorios Clinic, indicating a confirmed diagnosis of HIV; and (3) had received antiretroviral treatment for a minimum of one year so that adherence could be measured before and after implementation of the information system (HMC).

Measures and instruments

Participants were asked to complete a structured questionnaire that was designed by the researchers to collect information focused on socioeconomic status, clinical condition, and compliance with their prescribed care and medication regimen. The questionnaire collected information on various socioeconomic characteristics, including age, gender, sexual identity, marital status, level of educational attainment, annual income in the previous year, and substance use, including addictive substances and alcohol. Ten self-developed questions were introduced to measure the participants' compliance. These questions were designed to evaluate disease management, specifically, adherence to medication and scheduled appointments for laboratory tests. Each participant received an overall compliance score that ranged from 0 to 28. Participants with scores of 0 to 3 were classified as having good compliance. Those with scores of 4 to 6 were identified as moderately compliant, while scores of 7 or higher indicated poor compliance.

We also asked the participants to complete the WHO-QOL-HIV BREF questionnaire at the same time.

This questionnaire was developed by the World Health Organization (WHO) to measure the Health-Related Quality of Life (HRQOL) of HIV/AIDS patients. This questionnaire includes 31 items that cover six domains, including physical health (four items), psychological health (four items), level of independence (four items), social relationships (four items), environment (eight items), and spirituality (four items), along with two additional general items. The summary domain and total HRQOL scores were calculated using the scoring method developed by the WHOQOL-HIV group [5]. The researchers ensured the anonymity of the questionnaire by sealing it in an envelope. Upon receipt, they promptly coded the questionnaires to safeguard participant identities. Participants were randomly selected during their routine visits to the Grigorio Clinic at Larnaca General Hospital, minimizing potential bias.

Two hundred participants responded to the questionnaire at the study's outset. Subsequently, a sample of 34 patients was chosen for the six-month pilot implementation of the HMC information system. Subsequently, a sample of 34 patients was selected for the six-month pilot implementation of the HMC information system. The questionnaire was once again administered to this patient cohort, comprising the intervention group. Simultaneously, another group of 34 individuals without access to the information system completed the questionnaire, constituting the control group. To ensure data matching before and after the intervention, sociodemographic information was collected for each participant.

The HMC is an information system consisting of a Web Application (for medical and nursing staff) and an Android Application (for PLWH) where they interact in real time and aim to ensure the quality of life of HIV-positive people in Cyprus (Fig. 2). Specifically, the Android Application helps to improve the selfmonitoring of PLWH, through access to the results of their analyses, timely information about appointments, information regarding medical news around HIV and the Gregory Clinic, solving questions, sending messages with real-time response, etc. While the Web application allows medical and nursing staff to better track laboratory results, compliance with scheduled laboratory test appointments, prescription renewal and access to the patient's vaccination card. In addition, scheduling is automated according to international protocols for both HIV and syphilis, enabling medical and nursing staff to have two-way communication with each PLWH in real time. Moreover, the doctor can enter the prescription and the medical history (including the admission file). It is important to note that the app does not provide any information, including the phone number, email, Facebook account, messenger, or any details that would identify the person coded into the app. Both the Web App and the Android App require passwords for access, and each HIV-positive individual has exclusive access to their personal account. No HIV-positive individual is able to access another HIV-positive account, even if they know the password.

Fig. 2 Initial page of the developed Android and Web application (the Web application, mainly for the medical and nursing staff, is presented on the computer screen and the Android application, mainly for the PLWHA, is presented on the mobile screen)

Bias

To ensure impartiality, specific criteria for the exclusion of participants were established at the beginning of the study to minimize bias. Individuals aged \leq 18 years, not actively monitored at the Gregory Clinic and not receiving antiretroviral therapy for at least one year were specifically excluded.

Study size

Convenience sampling was considered the most suitable approach for data collection in this study, as it enabled us to include as many willing participants as possible more easily, quickly, and at a low cost, during a period of movement restrictions [6].

To meet the minimum sample requirements for all assessments, we distributed questionnaires to 200 patients before implementing the information system. Six months after the implementation of this system, 68 questionnaires were collected (34 from the implementation group and 34 from the controls), following a random selection from the original pool of 200 participants.

Statistical analysis

For descriptive statistical analysis, continuous variables were presented as mean values (mean, M) with calculated standard deviation (SD). Discrete variables were reported as frequencies (N) and relative frequencies (N%). The normality assumption was assessed using the Shapiro–Wilk test, as well as graphical representations, including the normal Q–Q plot, the detrended normal Q–Q plot, and the box plot.

The Wilcoxon signed-rank test was used to assess antiretroviral adherence outcomes in treated and untreated subjects with the HMC information system. To test the correlations between WHOQOL-HIV-BREF subscale scores and adherence level we used Kendall's tau_b.

Statistical analysis was performed using SPSS version 25 and STATA version 12 software. The minimum level of statistical significance was set at 5%. Analysis of Gain Scores were used to determine whether the intervention group exhibited significantly different pre-test to posttest scores compared to the control group.

Results

Sample characteristics

The initial study population included 200 HIV-positive patients (PLWHA) who were monitored at the Gregorios Clinic of Larnaca General Hospital in Cyprus. The mean time since diagnosis was 9.9 years (standard deviation [SD], 7.8 years). As shown in Table 1, the initial participant group included 142 males (71%) and 58 females

Table 1 Participant characteristics (N = 200)

	N	%
Sex		
Men	142	71.0
Women	58	29.0
Age (years)		
18–30	29	14.5
31-40	72	36.0
41–50	44	22.0
> 50	55	27.5
Sexual orientation		
Heterosexual	74	37.0
Homosexual	94	47.0
Bisexual	26	13.0
Transgender	1	0.5
Other	5	2.5
Educational level		
Primary education	36	18.0
Secondary education	140	70.0
Higher education	24	12.0
Marital status		
Married	48	24.0
Divorced	30	15.0
Widowed	14	7.0
Unmarried	108	54.0
Annual income		
No income	32	16.0
< 10,000 €	56	28.0
10,001-15,000 €	32	16.0
15,001-20,000 €	37	18.5
20,001-30,000 €	26	13.0
> 30,000 €	17	8.5
Nationality		
Greek	156	78.0
Other	44	22.0
Use of addictive substances		
No	149	74.5
Yes	51	25.5
Alcohol consumption		
Never	121	60.5
Rarely	34	17.0
Weekly	33	16.5
Daily	12	6.0
Comorbidities		
No	117	58.5
Yes	83	41.5
Do you know the medication you are taking?		
No	92	46.0
Yes	108	54.0
Does the treatment you are receiving seem to be I	nelping you?	
No	12	6.0

Table 1 (continued)

	Ν	%
Yes	188	94.0

(29%). Of this cohort, 14.5% were 18-30 years of age, 36% were 31-40 years of age, 22% were 41-50 years of age, and 27.5% were>50 years old. Seventy percent of these participants reported having a secondary education, while 18% had a primary education only and 12% reported tertiary education. Fifty-four percent of the participants were single, 24% were married, 15% were divorced, and 7% were widowed. Similarly, 28% of patients reported having no income while 16% reported income levels < €10,000 per year. Others (18.5%) reported an annual income between €10,001 and €15,000 per year. 18.5% of these people reported an annual income between 15,001 and 20,000 euros. While another 13% reported an annual income between €20,001 and €30,000, only 8.5% reported an income exceeding €30,000 per year. This patient cohort included 78% Greek natives and 22% who were foreigners in Cyprus. Furthermore, 25.5% of participants reported illicit use of addictive substances. While 60.5% of the participants reported that they never consumed alcohol, of those who did, 17% reported that they did so only on rare occasions, 16.5% consumed alcohol weekly, and 6% consumed alcohol on a daily basis.

Finally, we note that 41.5% of reported comorbidities, and 54% reported that they were aware of the names of the medications they were taking. Most (94%) believed that the treatments that they were receiving were beneficial (Table 1).

Impact of the implementation of an information system on quality of life

To determine whether the implementation of the monitoring system had a significant impact on patient QOL, we examined changes from baseline WHOQOL-HIV-BREF subscale scores reported by participants who were managed for six months using this system compared to those from patients who received routine care. Of note, we observed no statistically significant differences in the mean baseline scores in the intervention compared to the control group. The mean (M) ± standard deviation (SD) of the baseline scores on the physical health subscale were 12.6 ± 3.5 and 12.3 ± 5 in the intervention and control groups, respectively, with t(67) = 0.292 and p = 0.771. The M±SD of the baseline scores on the psychological health subscale were 12.7 ± 4.3 and 13.6 ± 4.2 in the intervention and control groups, respectively, with t(67) = -0.892 and p = 0.375. The $M \pm SD$ of the baseline scores on the degree of autonomy subscale were 14.4 ± 4.1 and 15.3 ± 5.3 in the intervention and control groups, respectively, with t(67) = -0.759 and p = 0.451. The M±SD of the baseline scores on the social relations subscale were 12.8 ± 4.6 and 14 ± 4.5 in the intervention and control groups, respectively, with t(67) = -1.104 and p = 0.274. The M±SD of the baseline scores on the living conditions subscale were 13.5 ± 3.6 and 13.8 ± 3.9 in the intervention and control groups, respectively, with t(67) = -0.315 and p = 0.754. Finally, the M ± SD of the baseline scores on the spirituality/religiosity/personal beliefs subscale were 14 ± 4.2 and 14.3 ± 4.4 in the intervention and control groups, respectively, with t(67) = -0.276 and p = 0.783 (Table 2; Fig. 3).

Table 2	Mean s	subscale scor	es on the two	o measures	by patient group
---------	--------	---------------	---------------	------------	------------------

Group	WHOQOL-HIV-BREF subscales	1st measurement		2nd measurement	
		Mean	SD	Mean	SD
Intervention (n=34)	Physical health	12.6	3.5	17.7	1.6
	Psychological health	12.7	4.3	15.7	2.9
	Degree of autonomy	14.4	4.1	16.6	3.3
	Social relationships	12.8	4.6	15.5	3.5
	Life conditions	13.5	3.6	17.1	1.4
	Spirituality/religious/personal Beliefs	14.0	4.2	16.7	3.6
Control (n = 34)	Physical health	12.3	5.0	12.9	4.7
	Psychological health	13.6	4.2	11.7	4.4
	Degree of autonomy	15.3	5.3	15.2	4.8
	Social relationships	14.0	4.5	12.8	5.1
	Life conditions	13.8	3.9	13.3	4.1
	Spirituality/religious/personal beliefs	14.3	4.4	14.0	4.6

Fig. 3 Perceptions of Quality of Life at Baseline Measurement and after six-month-follow-up by patient group

Comparisons between the intervention and control groups The mean changes in the scores (baseline to six-month follow-up) exhibited by participants in the intervention and control groups were assessed for each subscale of the WHOQOL-HIV-BREF using the Analysis of Gain Scores and independent samples *t*-tests based on the assumption that the data were normally distributed. The results of the independent samples *t*-tests revealed that the differences in baseline-to-six-month scores reported for the intervention group were significantly different from those reported for participants in the control group across all six subscales assessed by the WHOQOL-HIV-BREF questionnaire. Specifically, participants managed with the HMC information system responded with significant improvements in all six categories, including physical health, psychological health, degree of autonomy, social relations, living conditions, and spirituality/religion/personal beliefs.

Mean differences (assessed by subtracting the baseline from the six-month measurement for each subscale) were determined for participants in the intervention and the control groups. The results revealed several statistically significant differences between the two groups. We found that participants in the intervention group exhibited significantly higher mean score differences ($M \pm SD$) on the physical health subscale compared to those in the control group, at 5.1 ± 3 versus 0.6 ± 6.7 , respectively, with t(46,234) = 3.564 and p = 0.001. Participants in the intervention group also exhibited significantly higher mean score differences on the psychological health subscale compared to those in the control group, at 3±3.3 ver $sus - 2.1 \pm 5$, respectively, with t(66) = 4.993 and *p* < 0.001. Similar results were observed for other subscales. For example, participants in the intervention group exhibited significantly higher mean score differences on the degree of autonomy subscale compared to those in the control group, at 2.2 ± 4.7 versus -0.4 ± 5.3 , respectively, with t(66) = 2.113 and p = 0.038. Participants in the intervention group exhibited significantly higher mean score differences on the social relations subscale compared to those in the control group, at 2.7 ± 5.4 versus -1.1 ± 6.1 , respectively, with t(66) = 2.749 and p = 0.008. Similarly, participants in the intervention group exhibited significantly higher mean score differences on the life conditions subscale compared to those in the control group, at 3.6 ± 3.4 versus - 0.7 ± 4.9, respectively, with t(66) = 4.152 and p < 0.001. Finally, participants in the intervention group exhibited significantly higher mean score differences on the spirituality/religious/personal beliefs subscale compared to those in the control group, at 2.8 ± 4.4 *versus* -0.2 ± 5.5 , respectively, with t(66) = 2.456 and p = 0.017. Table 2 provides the mean scores for the two measurements, along with the corresponding standard deviations, for each group of patients (intervention and control).

Adherence to antiretroviral therapy and quality of life

As shown in Table 3, we found that higher WHOQOL-HIV-BREF scores overall were associated with better patient adherence across all subscales. Specifically, a detailed analysis of the results shown in Table 3 revealed negative significant associations between subscale scores and patient adherence. For example, our findings revealed a statistically significant negative relationship between scores on the physical health subscale and the degree of medical compliance $(\tau b(200) = -0.177, p = 0.002)$ or on the psychological health subscale and the degree of compliance $(\tau b(200) = -0.262, p < 0.001)$. Furthermore, our findings revealed a statistically significant negative correlation between the scores on the degree of autonomy subscale and the degree of compliance $(\tau b(200) = -0.208)$, p = 0.001), on the social relations subscale and the level of compliance $(\tau b(200) = -0.264, p < 0.001)$, on the life conditions subscale and the degree of compliance $(\tau b(200) = -0.224, p < 0.001)$, and on the spirituality/ religious/personal beliefs subscale $(\tau b(200) = -0.262)$, p < 0.001) that correlate with better adherence. This result suggests that higher scores on the physical health, the psychological health, on the degree of autonomy, on the social relations, on the life conditions, and on the

Table 3 Tests of correlations between WHOQOL-HIV-BREF subscale scores and adherence level (N = 200)

 Kendall's tau_b	
Physical health	
Correlation coefficient	177**
Sig. (2-tailed)	.002
Psychological health	
Correlation coefficient	262**
Sig. (2-tailed)	.000
Degree of autonomy	
Correlation coefficient	208**
Sig. (2-tailed)	.001
Social relationships	
Correlation coefficient	264**
Sig. (2-tailed)	.000
Life conditions	
Correlation coefficient	224**
Sig. (2-tailed)	.000
Spirituality/religious/personal beliefs	
Correlation coefficient	262**
Sig. (2-tailed)	.000
N	200

spirituality/religious/personal beliefs subscale are linked to better adherence (Table 3). Figure 4 further illustrates the correlation between the degree of adherence to antiretroviral treatment and the quality of life. Notably, the data indicate that higher scores across all subscales are associated with better treatment compliance.

Adherence to antiretroviral therapy in individuals who were and were not managed with a monitoring system

The Wilcoxon signed-rank test revealed a statistically significant improvement in medical adherence (z=-3.819, p<0.001) for those managed with a monitoring system. By contrast, no statistically significant improvements in adherence were observed (z=-1.414, p=0.157) in subjects who did not have the monitoring system in place (Table 4). Interestingly, no significant associations were found between compliance and marital status, economic status, ethnicity, years since diagnosis, or specific comorbidities.

Discussion

The association between adherence to ART and improved QOL has been demonstrated consistently in previous research studies [7–19]. This finding was further supported by the findings presented in this study that revealed a positive correlation between improved compliance and higher scores in various domains of QOL, including physical health, psychological health, degree of autonomy, social relations, life conditions, and spiritual-ity/religious/personal beliefs.

Our findings can be compared to those reported by Silva et al. [7], who found that PLWHA in Brazil who did not adhere to the ART regimen received the lowest scores in all domains of QOL. Among their conclusions, these researchers identified ART adherence as a positive factor related to QOL because of its beneficial effects on immunity, control of viral load, and disease progression.

It is critical to emphasize that achieving virus suppression and reducing viral load in HIV-infected individuals requires not only strong adherence to ART but to appropriate medical treatment [8, 20–23]. Results from several previous studies revealed that PLWHA who receive appropriate medical treatment will have a life expectancy similar to that experienced by the general population [24, 25]. However, achieving this near-normal life expectancy is contingent upon at least 95% adherence to ART to prevent treatment failure and the development of drug resistance [8, 26, 27]. Thus, the evaluation of new therapeutic options and technologies aimed at improving healthcare for PLWHA should include assessments of HRQOL in addition to clinical endpoints such as CD4⁺ T-cell count, viral load, and disease progression [24].

Fig. 4 Correlation between the quality of life and adherence to antiretroviral treatment

Group	Degree of compliance (1st measurement)			Degree of compliance (2nd measurement)		
	Good compliance	Moderate compliance	Poor compliance	Good compliance	Moderate compliance	Poor compliance
Intervention						
Count	13	14	7	26	6	2
Row N %	38.2%	41.2%	20.6%	76.5%	17.6%	5.9%
Control						
Count	18	4	12	18	6	10
Row N %	52.9%	11.8%	35.3%	52.9%	17.6%	29.4%

Table 4	Allocation	of patients	by aroup	and	measurement
	/ mocurion	or patients	o, gioup	ana	incusure incrit

A research study conducted in Baghdad by Imam et al. [27] highlighted a strong relationship between adherence to a highly-active ART (HAART) regimen and quality of life (QOL). The findings presented in this study revealed that patients with only low-to-moderate adherence to HAART were 60% less likely to present with high global QOL scores compared to patients with high adherence to this regimen. As stated by Tsiadis [28], p. 33], "poor adherence leads to significant morbidity, mortality, and avoidable health care costs." He also noted that while

technology can be a valuable tool in improving patient compliance, these devices are not stand-alone solutions to this problem.

Mobile devices, particularly those capable of transmitting and receiving short message service (SMS) communications (i.e., texts), have become increasingly prevalent in HIV/AIDS clinics as a means to support interactions with patients. In 2010, Lester et al. [29] conducted the first randomized clinical trial that explored the use of cellular devices to monitor the adherence of HIV-positive patients to first-line ART. The results from this study revealed that the use of cellular devices led to a modest improvement in self-reported adherence and viral suppression. Likewise, results from a study published by Luque et al. [30] revealed a significant increase in selfefficacy among study participants who utilized the personal health record application.

In recent years, there has been a notable rise in the development and implementation of mobile health (mHealth)-based self-management interventions for PLWHA. These interventions were demonstrated to be feasible, acceptable, and effective when used by study participants [24]. As stated by Muessig [31], p. 15], recent mHealth interventional tools designed to support medication adherence "show acceptability, feasibility, and preliminary efficacy in increasing ART use intentions, self-reported adherence, and viral suppression". Similarly, in a literature review published by Cooper et al. [24], the authors concluded that mHealth interventions can have a significant impact on various outcomes related to HIV care. These outcomes include ART adherence [32-39], suppression of viral load [34, 37, 38], engagement with care [40], HIV knowledge, depression, anxiety, and selfefficacy [31].

Many mHealth interventions provide patients with a wide range of functionality. One key aspect is their ability to monitor clinical indicators that reflect the symptoms and progression of the disease.

Furthermore, mHealth interventions facilitate selfmanagement by enabling patients to monitor and control their diet and exercise routines and provide access to information that is tailored to specific guidelines and recommendations for managing HIV. Patients can also receive educational messages and electronic support from their peers that address their unique needs and preferences [24]. Many mHealth interventions also offer a high degree of flexibility and facilitate the delivery of customized content that can be tailored to the specific needs of each user [41]. The accessibility of mHealth interventions helps patients to overcome various social and structural barriers [42]. Furthermore, mHealth interventions have demonstrated great potential for scalability and diffusion across different geographic locations and may be particularly useful in resource-constrained settings [43].

The present study demonstrated that the implementation of an Android-application-based monitoring system led to improved adherence among those who participated in the intervention compared to those who did not. Additionally, we found that the implementation of this information system had positive effects on physical health, mental health, degree of autonomy, social relationships, and living conditions. These findings align with results published by Mehraeen et al. [43] which also highlighted the significant positive impact of mHealth strategies on disease prevention, ART adherence, and treatment for PLWHA. Furthermore, mHealth interventions had positive effects on behavioral and social problems that typically affect PLWHA. While the mHealth market needs to be regulated, we recognize the overall importance of mHealth for self-monitoring, self-care, and self-management by PLWHA [43].

However, despite the demonstrated benefits of mHealth interventions, there is a significant delay in their application in the clinical field, particularly in Cyprus. One of the major challenges faced by healthcare providers worldwide is securing the necessary financial resources to support the implementation of these critical interventions.

Conclusions

The findings presented in this study highlight the positive impact of an HMC information system on QOL and medical adherence of PLWHA. The implementation of this system resulted in improvements in physical health, mental health, degree of autonomy, social relationships, and living conditions. Compliance with medical care and drug therapy were also positively associated with higher scores in various subscales on the WHOQOL-HIV-BREF questionnaire, including physical health, psychological health, degree of autonomy, social relationships, life circumstances, and spirituality/religious/personal beliefs. Our study also identified specific demographic factors associated with increased compliance. Male patients, those who are elderly, those with higher levels of educational attainment, individuals who do not use addictive substances, and those who utilized the monitoring information system were significantly more compliant with medical care and drug therapy. Overall, our findings indicate that managing PLWHA with an information system can contribute significantly to an improved QOL as well as adherence to life-saving medical treatment.

Abbreviations

 HMC
 Health Medical Care

 ART
 Antiretroviral therapy

 HIV
 Human immunodeficiency virus

 AIDS
 Acquired immune deficiency syndrome

 WHO
 World Health Organization

PLWHA People living with HIV/AIDS

Acknowledgements

The authors would like to thank all the research participants, the Pancyprian Association of Nurses and Midwives (PASYNM), and the AIDS Support Movement in Cyprus for their support.

Author contributions

All authors contributed to the study conception and design. Material preparation, data collection and analysis were performed by Maria Panayi. The first draft of the manuscript was written by Maria Panayi and all authors

commented on previous versions of the manuscript. All authors read and approved the final manuscript.

Funding

This work was supported by the Pancyprian Association of Nurses and Midwives (PASYNM), and the AIDS Support Movement in Cyprus. Author M.P. has received research support from the State Scholarship Foundation of Cyprus (IKYK) and the Pancyprian Association of Nurses and Midwives (PASYNM).

Availability of data and materials

The dataset used and analyzed in this study is available from the corresponding author upon reasonable request.

Declarations

Ethics approval

This study was performed in line with the principles of the Declaration of Helsinki. Approval was granted by the National Bioethics Committee of Cyprus and by the Scientific Promotion Committee OKYPY (Organization Public Health Services) on April 23 and July 15, 2020, respectively.

Consent to participate

All participants provided written informed consent.

Consent to publish

Not applicable.

Competing interests

The authors have no relevant financial or non-financial interests to disclose.

Author details

¹Ph.D. Programme Health Management, Frederick University, Nicosia, Cyprus. ²Gregorios AIDS Clinic, General Hospital of Larnaca, Larnaca, Cyprus. ³Hippocration General Hospital, Athens, Greece. ⁴Department of Public Health, School of Medicine, University of Patras, Patras, Greece.

Received: 17 October 2023 Accepted: 21 December 2023 Published online: 23 January 2024

References

- 1. UNAIDS (2021) Global HIV and AIDS statistics. https://www.unaids.org/ en/resources/fact-sheet. Accessed 20 Dec 2022
- Joint United Nations Programme On HIV/AIDS (2014) Fast-track: ending the AIDS epidemic by 2030. UNAIDS, Geneva, pp 1–40. https://www. unaids.org/sites/default/files/media_asset/JC2686_WAD2014report_ en.pdf. Accessed 20 Dec 2022
- UNAIDS (2022) Global HIV and AIDS statistics. https://www.unaids.org/ sites/default/files/media_asset/UNAIDS_FactSheet_en.pdf. Accessed 18 Nov 2023
- HIV/AIDS (2022) Epidemiological Bulletin Period from 1986 to 2021. Ministry of Health Cyprus, pp 1–8
- WHOQOL HIV Group (2004) WHOQOL-HIV for quality of life assessment among people living with HIV and AIDS: results from the field test. AIDS Care 16(7):882–889.https://doi.org/10.1080/095401204123312 90194
- Thomeas D (2022) Sampling methods. Thesis. School of Economic Sciences and Business Administration Department of Tourism Administration, University of Patras, pp 1–72 (in Greek). http://repository.library. teiwest.gr/xmlui/bitstream/handle/123456789/10558/LOG-%ce%98% ce%a9%ce%9c%ce%95%ce%91%ce%a3%20%ce%94%ce%97%ce%92% ce%97%ce%a4%ce%91%ce%93%ce%94%ce%94%ce%97%ce%92% ce%97%ce%a4%ce%91%ce%99%ce%94%ce%94%ce%97%ce%92% ce%97%ce%a4%ce%91%ce%99%ce%94%ce%94%ce%94%ce%97%ce%92%
- e Silva ACO, Reis RK, Nogueira JA, Gir E (2014) Quality of life, clinical characteristics, and treatment adherence of people living with HIV/AIDS. Rev Latino-Americana Enferm 22(6):994–1000. https://doi.org/10.1590/ 0104-1169.3534.2508

- Galvão MTG, Soares LL, Pedrosa SC, Fiuza MLT, Lemos LA (2015) Quality of life and adherence to antiretroviral medication in people with HIV. Acta Paul Enferm 28(1):48–53. https://doi.org/10.1590/1982-0194201500009
- Luszczynska A, Sarkar Y, Knoll N (2007) Receiving social support, selfefficacy, and finding benefits in disease as predictors of physical functioning and adherence to antiretroviral therapy. Patient Educ Counsel 66(1):37–42. https://doi.org/10.1016/j.pec.2006.10.002
- Mannheimer SB, Matts J, Telzak E, Chesney M, Child C, Wu AW, Friedland G (2005) Quality of life in HIV-infected individuals receiving antiretroviral therapy is related to adherence. AIDS Care 17(1):10–22. https://doi.org/10. 1080/09540120412331305098
- Parsons TD, Braaten AJ, Hall CD, Robertson KR (2006) Better quality of life with neuropsychological improvement on HAART. Health Qual Life Outcomes 4:11. https://doi.org/10.1186/1477-7525-4-11
- 12. Reis A, Guerra MP, Lencastre L (2013) Treatment adherence, quality of life and clinical variables in HIV/AIDS infection. World J AIDS 3(3):239–250. https://doi.org/10.4236/wja.2013.33032
- Liping M, Peng X, Haijiang L, Lahong J, Fan L (2015) Quality of life of people living with HIV/AIDS: a cross-sectional study in Zhejiang Province, China. PLoS ONE 10(8):e0135705. https://doi.org/10.1371/journal.pone. 0135705
- Ruiz-Pérez I, Olry de Labry-Lima A, López-Ruz MA, del Arco-Jiménez A, Rodríguez-Baño J, Causse-Prados M, Pasquau-Liaño J, Martín-Rico P, Prada-Pardal JL, de la Torre-Lima J, López-Gómez M, Marcos M, Muñoz N, Morales D, Muñoz I (2005) Clinical status, adherence to HAART and quality of life in HIV-infected patients receiving antiretroviral treatment. Enferm Infecc Microbiol Clin 23(10):581–5. https://doi.org/10.1157/13081 565
- Sherr L, Lampe F, Norwood S (2007) Successive switching of antiretroviral therapy is associated with a high psychological and physical burden. Inter J STD AIDS 18(10):700–704. https://doi.org/10.1258/0956462077 82193821
- Desta A, Biru TT, Kefale AT (2020) Health-related quality of life of people receiving highly active antiretroviral therapy in Southwest Ethiopia. PLoS ONE 15:1–15. https://doi.org/10.1371/journal.pone.0237013
- Nakagawa F, Lodwick RK, Smith CJ, Smith R, Cambiano V, Lundgren JD, Delpech V, Phillips AN (2012) Projected life expectancy of people with HIV according to the timing of diagnosis. AIDS 26(3):335–343. https://doi.org/ 10.1097/QAD.0b013e32834dcec9
- Beyene K, Engidawork E (2010) Quality of life of people living with HIV/ AIDS and on highly active quality of life of people living with HIV/AIDS and on highly active antiretroviral therapy in Ethiopia. Afr J AIDS Res 9(1):31–40. https://doi.org/10.2989/16085906.2010.484560
- Imam MH, Ferdous C, Akhter S (2011) Health-related quality of life among the people living with HIV. Bangladesh Med Res Counc Bull 37:1–6. https://doi.org/10.3329/bmrcb.v37i1.7791
- García PR, Côté JK (2003) Factors affecting adherence to antiretroviral therapy in people living with HIV/AIDS. J Assoc Nurs AIDS Care 14(4):37– 45. https://doi.org/10.1177/1055329003252424
- Huang L, Quartin A, Jones D, Havlir DV (2006) Intensive care of patients with HIV infection. N Engl J Med 355(2):173–181. https://doi.org/10.1056/ NEJMra050836
- Laurence J, Hodtsev AS, Posnett DN (1992) Superantigen implicated in dependence of HIV-1 replication in T cells on TCR V beta expression. Nature 358(6383):255–259. https://doi.org/10.1038/358255a0
- 23. Mweemba P, Makukula MK, Mukwato PK, Makoleka MM (2010) Quality of life and adherence to antiretroviral drugs. Med J Zambia 37(1):31–9
- Cooper V, Clatworthy J, Whetham J et al (2017) mHealth interventions to support self-management in HIV: a systematic review. Open AIDS J 11:119132. https://doi.org/10.2174/1874613601711010119
- 25. Dimoni C (2008) The effect of symptoms and drug side effects on the perceptions and compliance to antiretroviral therapy of HIV/AIDS patients. Doctoral dissertation, National and Kapodistrian University of Athens. National Archive of Ph.D., pp 1–188. http://hdl.handle.net/10442/ hedi/18410. Accessed 20 Dec 2022
- Wilson IB, Cleary PD (1995) Linking clinical variables with health-related quality of life: a conceptual model of patient outcomes. JAMA 273(1):59– 65. https://doi.org/10.1001/jama.1995.03520250075037
- Oguntibeju OO (2012) Quality of life of people living with HIV and AIDS and antiretroviral therapy. HIV AIDS 4:117–124. https://doi.org/10.2147/ HIV.S32321

- Lester RT, Ritvo P, Mills EJ, Kariri A, Karanja S, Chung MH, Jack W, Habyarimana J, Sadatsafavi M, Najafzadeh M, Marra CA, Estambale B, Ngugi E, Ball TB, Thabane L, Gelmon LJ, Kimani J, Ackers M, Plummer FA (2010) Effects of a mobile phone short message service on antiretroviral treatment adherence in Kenya (WelTel Kenya1): a randomized trial. Lancet 376(9755):1838–1845. https://doi.org/10.1016/S0140-6736(10)61997-6
- Luque AE, Corales R, Fowler RJ, DiMarco J, van Keken A, Winters P, Keefer MC, Fiscella K (2013) Bridging the digital divide in HIV care: a pilot study of an iPod personal health record. J Int Assoc Provid AIDS Care 12(2):117–121. https://doi.org/10.1177/1545109712457712
- Moore DJ, Poquette A, Casaletto KB et al (2015) Individualized texting for adherence building (iTAB): improving antiretroviral dose timing among HIV-infected persons with co-occurring bipolar disorder. AIDS Behav 19(3):459–471. https://doi.org/10.1007/s10461-014-0971-0
- Muessig KE, LeGrand S, Horvath KJ, Bauermeister JA, Hightow-Weidman LB (2017) Recent mobile health interventions to support medication adherence among HIV-positive MSM. Curr Opin HIV AIDS 12(5):432–441. https://doi.org/10.1097/COH.00000000000401
- Sabin LL, Bachman DeSilva M, Gill CJ et al (2015) Improving adherence to antiretroviral therapy with triggered real-time text message reminders: the China adherence through technology study. J Acquir Immune Defic Syndr 69(5):551–559. https://doi.org/10.1097/QAI.000000000000651
- Murray M (2015) mHEALTH to Improve Health: Effectiveness of a weekly text messaging intervention to improve ART adherence and HIV Viral Load: WeITel OAKTREE
- 34. Dowshen N, Kuhns LM, Johnson A, Holoyda BJ, Garofalo R (2012) Improving adherence to antiretroviral therapy for youth living with HIV/AIDS: a pilot study using personalized, interactive, daily text message reminders. J Med Internet Res 14(2):e51. https://doi.org/10.2196/jmir.2015
- Hailey JH, Arscott J (2013) Using technology to effectively engage adolescents and young adults into care: STAR TRACK Adherence Program. J Assoc Nurs AIDS Care 24(6):582–586. https://doi.org/10.1016/j.jana.2013. 03.001
- Lewis MA, Uhrig JD, Bann CM et al (2013) Tailored text messaging intervention for HIV adherence: a proof-of-concept study. Health Psychol 32(3):248–253. https://doi.org/10.1037/a0028109
- Perera AI, Thomas MG, Moore JO, Faasse K, Petrie KJ (2014) Effect of a smartphone application incorporating personalized health-related imagery on adherence to antiretroviral therapy: a randomized clinical trial. AIDS Patient Care STDS 28(11):579–586. https://doi.org/10.1089/apc. 2014.0156
- Swendeman D, Jana S, Ray P, Mindry D, Das M, Bhakta B (2015) Development and pilot testing of daily interactive voice response (IVR) calls to support antiretroviral adherence in India: a mixed-methods pilot study. AIDS Behav 19(Suppl. 2):142–155. https://doi.org/10.1007/ s10461-014-0983-9
- Siedner M, Santorino D, Lankowski A, Kanyesigye M, Bwana M, Haberer J et al (2014) An SMS intervention to improve HIV linkage to care: a randomized, comparative effectiveness trial. In: Conference on retroviruses and opportunistic infections, Boston
- Uhrig JD, Lewis MA, Bann CM, Harris JL, Furberg RD, Coomes CM et al (2012) Addressing HIV knowledge, risk reduction, social support, and patient involvement using SMS: results of a proof-of-concept study. J Health Commun Int Perspect 17:128–145. https://doi.org/10.1080/10810 730.2011.649156
- Amico KR (2015) Evidence for technology interventions to promote ART adherence in adult populations: a review of the literature 2012–2015. Curr HIV/AIDS Rep 12(4):441–450. https://doi.org/10.1007/ s11904-015-0286-4
- Haberer JE, Sabin L, Amico KR et al (2017) Improving antiretroviral therapy adherence in resource-limited settings at scale: a discussion of interventions and recommendations. J Int AIDS Soc 20(1):21371. https://doi.org/ 10.7448/IAS.20.1.21371
- Mehraeen E, SeyedAlinaghi S, Pashaei Z et al (2022) Mobile applications in HIV self-management: a systematic review of scientific literature. AIDS Rev 24(1):24–31. https://doi.org/10.24875/AIDSRev.21000025

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Submit your manuscript to a SpringerOpen[®] journal and benefit from:

- Convenient online submission
- Rigorous peer review
- Open access: articles freely available online
- High visibility within the field
- Retaining the copyright to your article

Submit your next manuscript at > springeropen.com